Author's response to reviews

Title: An observational study on Awareness Regarding Eye Donation among Stakeholders in Srikakulam District in South India

Authors:

Venkata Ramana Ronanki (ramana.ronanki@gmail.com)
Sheeladevi Sethu (sheela@lvpei.org)
Brinda P Ramachandran (bru_darsini@yahoo.org)
Isabelle Jalbert (i.jalbert@unsw.edu.au)

Version: 2 Date: 8 April 2013

Author's response to reviews: see over
Dear Editor

Please find my response to your queries raised

Editorial Requests:

1. Competing interests:

Manuscripts should include a 'Competing interests' section. This should be placed after the Conclusions/Abbreviations. Please consider the following questions and include a declaration of competing interests in your manuscript:

Financial competing interests

? In the past five years have you received reimbursements, fees, funding, or salary from an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future? Is such an organization financing this manuscript (including the article-processing charge)? If so, please specify.

? Do you hold any stocks or shares in an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future? If so, please specify.

? Do you hold or are you currently applying for any patents relating to the content of the manuscript? Have you received reimbursements, fees, funding, or salary from an organization that holds or has applied for patents relating to the content of the manuscript? If so, please specify.

? Do you have any other financial competing interests? If so, please specify.

Non-financial competing interests

? Are there any non-financial competing interests (political, personal, religious, ideological, academic, intellectual, commercial or any other) to declare in relation to this manuscript? If so, please specify.

2. Authors' Contributions

In order to give appropriate credit to each author of a paper, the individual contributions of authors to the manuscript should be specified in this section.

An 'author' is generally considered to be someone who has made substantive intellectual contributions to a published study. To qualify as an author one should 1) have made substantial contributions to conception and design, or acquisition of data, or analysis and interpretation of data; 2) have been involved in drafting the manuscript or revising it critically for important intellectual content; and 3) have given final approval of the version to be published. Each author should have participated sufficiently in the work to take public responsibility for appropriate portions of the content. Acquisition of funding, collection of data, or general supervision of the research group, alone, does not justify authorship.

We suggest the following kind of format (please use initials to refer to each author's contribution): AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and
coordination and helped to draft the manuscript. All authors read and approved the final manuscript.

All contributors who do not meet the criteria for authorship should be listed in an acknowledgements section. Examples of those who might be acknowledged include a person who provided purely technical help, writing assistance, or a department chair who provided only general support.

3. Title page should be in the manuscript

The title page should:

- provide the title of the article
- list the full names, institutional addresses and email addresses for all authors
- indicate the corresponding author

Please note:

- the title should include the study design, for example "A versus B in the treatment of C: a randomized controlled trial X is a risk factor for Y: a case control study"
- abbreviations within the title should be avoided

4. Abstract should be in the manuscript
http://www.biomedcentral.com/bmcoptomol/authors/instructions/researcharticle#formatting-abstract

Please include an abstract as the second page in your manuscript file (directly following the title page). Please format your abstract according to the guidelines for authors http://www.biomedcentral.com/authors/authorfaq/format. Potential referees will be asked to review the manuscript having seen only the title and abstract, so it is important that these are both informative and concise.